Literatur

[Gus97] D. Gusfield: Algorithms on Strings, Trees and Sequences: Computer Science and Computational Biology, Cambridge University Press, Cambridge, 1997.

[Kur01] S. Kurtz: Foundations of Sequence Analysis, University of Bielefeld, July 2001, Lecture notes for a course in the Winter Semester 2000/2001.

[Pev00] P. A. Pevzner: Computational Molecular Biology, The MIT Press, Cambridge, 2000.

[Set97] J. Setubal, J. Meidanis: Introduction to Computational Molecular Biology, PWS Publishing, Boston, Mass., 1997.

[Wat95] M. S. Waterman: Introduction to Computational Biology: Maps, Sequences and Genomes, Chapman & Hall, London, 1995.